Author's response to reviews

Title: The Effect of Oriental Medicine Music Therapy on Hwa-byung Patients: study protocol for a randomized controlled trial

Authors:

Jeong-Su Park (suyahpark@gmail.com)
Sunju Park (sjpark422@khu.ac.kr)
Chun-Hoo Cheon (pm.thehoo@khu.ac.kr)
Sun-Yong Chung (lovepwr@khu.ac.kr)
Jong-Woo Kim (aromagi@khu.ac.kr)
Yong-Cheol Shin (syc99@khu.ac.kr)
Seong-Gyu Ko (epiko@khu.ac.kr)

Version: 5 Date: 10 August 2012

Author's response to reviews: see over
August 10th, 2012

Re: Manuscript #1001460512663772

Dear Editor,

Thank you very much for your acceptance decision on our manuscript—The Effect of Oriental Medicine Music Therapy on Hwa-byung Patient: study protocol for a randomized controlled trial for Trials Journal (#1001460512663772).

The revised version of the manuscript is included along with this letter for publication in Trials Journal. We have corrected some minor issues on the manuscript and the figures.

Thank you very much for giving us a chance to publish our work in your journal.

Yours sincerely,

Seong-Gyu Ko, M.D., Ph.D., M.P.H.

Professor
Center for Clinical Research and Genomics,
Kyung Hee University.
1, Hoegi-dong, Dongdaemun-gu,
Seoul, 130-701, Korea

(Email) epiko@khu.ac.kr
(Phone) 82-2-961-0329
(Fax) 82-2-966-1165
Major compulsory revisions

1) Can you please also ensure that your study protocol is formatted as outlined in our instructions for authors? Please make sure to include all sections listed and to order them correctly. At present the trial status section is missing. Please note that figures and tables should not be included in the main body of the manuscript please see the previous link for more information on how to include them.

We checked all the details of your instructions for authors. The keywords section was added. The trial status section was described in the manuscript. The figures and tables are not included in the main body.